All Exp Mtg
February 1, 2010

NU.MI MINOS Status Report Gregory Pawloski
NEAR DETECTOR:

Uptime last week:
99.20% (P.O.T. Weighted)

Issues from last week
No major issues

Recalibrated hot channel
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NuMI-MINOS Status Report — Gregory Pawloski
FAR DETECTOR:

Uptime last week:
99.75% (P.O.T. Weighted)

Issues from last week

Last Monday HV system tripped but auto-recovered
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NuMI-MINOS Status Report — Gregory Pawloski

NuMI:

Shutdown Work
Complete PM on MI65 dehumidification system
Complete PM on Target Hall Chiller
MI65 glycol make up system operational
Replace LCW temperature control valve on Horn 2 RAW Skid

Replace humidity sensor on target pile air system

Complete water samples and flow rates in Pre-Target, Decay Pipe,
and Absorber Hall drains

Finish upgrades on AH dehumidification controls to allow for full
remote operational control

Replace damaged heater controller on AH Munter unit #2

Replace heat exchanger on AH dehumidification system
Current Issues:

Unit B dehumidifer (1 of 2 units) tripped on Saturday

Currently investigating
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ry Pawloski

NuMI:

4.2e18 POT collected last week

Week ending 00:00 Monday 01 February 2010
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Protons on Target for FY10
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1.2e20 integrated protons on target for FY10/RHC
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Protons on Target for NuMI
Total NuMI protons to 00:00 Monday 01 February 2010
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